Author’s response to reviews

Title: Rotation of a toric intraocular lens with and without capsular tension ring: data from a multicenter non-inferiority randomized clinical trial (RCT)

Authors:

Ursula Hahn (Ursula.Hahn@uni-wh.de)
Frank Krummenauer (Frank.Krummenauer@uni-wh.de)
Stefanie Schmickler (st.schmickler@augen-zentrum-nordwest.de)
Jörg Koch (joerg.koch@augen-franziskus.de)

Version: 5 Date: 15 Jun 2019

Author’s response to reviews:

1. Trial registration
   We note you have provided your ethics approval information in the trial registration section rather the Trial registration. The Trial registration section should include the Trial registration number (previously stated as: DRKS00015316) as well as the name of the trial registry it is registered in i.e. the DRKS. It should also include the date of registration and a statement that the trial was registered retrospectively.

   Authors: Hopefully I did it right this time. I added the information about the retrospective trial registration in the abstract and in the declaration section. As far as the duplicate entry is not required, I would be grateful you would delete the information in the place where it is not required.

2. Author contributions.
   Please include the statement that author JK approves the final submission in the Author contributions section as has been done for all other authors.

   Authors: Hopefully I did it right this time: I added the words „given final approval” in the Author contribution section to the entry of Jörg Koch (JK)

3. Clean manuscript
   At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
   Please remove any files that should not be published along with your manuscript e.g. cover letters, reviewer responses, guidelines etc., ready for publication.
Authors: We uploaded a single clean copy of the manuscript.